Author's response to reviews

Title: Quality of Care Associated with Number of Cases Seen and Self-reports of Clinical Competence for Japanese Physicians-in-training in Internal Medicine

Authors:

Yasuaki Hayashino (hayasino@kuhp.kyoto-u.ac.jp)
Shunich Fukuhara (fukuhara@phb.med.kyoto-u.ac.jp)
Kunihiko Matsui (kmatsui@kaiju.med.kumamoto-u.ac.jp)
Yoshinori Noguchi (yoshi-noguchi@umin.ac.jp)
Taro Minami (nantaro@gmail.com)
Dan Bertenthal (dbertenthal@itsa.ucsf.edu)
John W Peabody (peabody@psg.ucsf.edu)
Yoshitomo Mutoh (ym6104@toranomon.gr.jp)
Yoshihiko Hirao (hirao@naramed-u.ac.jp)
Kazuhiko Kikawa (kikawa@fc.kuh.kumamoto-u.ac.jp)
Yohei Fukumoto (yoheifuk@yamaguchi-u.ac.jp)
Junichiro Hayano (hayano@med.nagoya-cu.ac.jp)
Teruo Ino (ino@fujita-hu.ac.jp)
Umihiko Sawada (usawada@med.nihon-u.ac.jp)
Jin Seino (S7919@snh.go.jp)
Norio Higuma (nhiguma@hosp.niigata.niigata.jp)
Hiroyasu Ishimaru (ishi-hir@mx5.nisiq.net)

Version: 4  Date: 6 June 2006

Author's response to reviews:

June 4th, 2006
The Editors
BMC Medical Education

Dear Editors,

Electrically submitted is our revised manuscript "Quality of Care Associated with Number of Cases Seen and Self-reports of Clinical Competence for Japanese Physicians-in-training in Internal Medicine" We revised our manuscript considering helpful comments from a reviewer, and would like to resubmit this paper for your consideration for BMC Medical Education. Our responses to the reviewer's comments was uploaded as MS Word file named as "responses.doc". A major point indicated by the reviewer was about the flow of the discussion and grammatical error, but our manuscript was edited by a native English user, actually a Professor at a medical school in the United States. Despite of that, we had other native English user check the flow and grammatical error of our manuscript again. The reviewer's comment was vague and not specific, and we would be glad if the reviewer may specify the points that should be revised if further revise is needed.

I hereby certify that there has been no duplicate publication or submission elsewhere of any part of the work in any language. All authors meet criteria of authorship, will sign a statement attesting to authorship, disclosing all potential conflicts of interest, and releasing the copyright should the manuscript be accepted for publication. We have no affiliation with any organization with a financial interest. One of authors, Fukuhara Shunich was supported in part by a grant-in-aid from the Ministry of Health, Labor, and Welfare of Japan.

Any correspondence should be directed to Yasuaki Hayashino at the following address. Thank you for attention to our paper.

Sincerely yours

Yasuaki Hayashino, MD, MPH
Department of Epidemiology and Healthcare Research Kyoto University
Graduate School of Medicine Kone-cho, Japan
Yoshida, Saky-ku, Kyoto 606-8501
Phone 075-753-9467
FAX 075-753-4644
hayasino@kuhp.kyoto-u.ac.jp